Author’s response to reviews

Title: A systematic review and meta-analysis of the prevalence of thrombosis and bleeding at diagnosis of Philadelphia-negative myeloproliferative neoplasms

Authors:

Tarinee Rungjirajittranon (taitharee@gmail.com)
Weerapat Owattanapanich (weerapato36733@gmail.com)
Patompong Ungprasert (p.ungprasert@gmail.com)
Noppadol Siritanaratkul (sinoppadol@gmail.com)
Theera Ruchutrakool (truchutrakool@gmail.com)

Version: 2 Date: 10 Feb 2019

Author’s response to reviews:

BMC Cancer

11 February 2019

Dear Editor,

Manuscript number: BCAN-D-18-02581R1

Thank you for your letter and insightful comments concerning our manuscript titled “A systematic review and meta-analysis of the prevalence of thrombosis and bleeding at diagnosis of Philadelphia-negative myeloproliferative neoplasms: BCAN-D-18-02581R1”. We have studied your comments/suggestions carefully and have revised our manuscript accordingly (detailed below).

Response to editor:

Comment 1: Please ensure that 'Table 1' has been uploaded with your resubmission.
Reply: We have to apologize about the mistake. We have now uploaded Table file already.

Comment 2: We note that two authors have the same initials (TR). To distinguish between them in the Authors’ Contributions section please designate them as TR1 and TR2, with TR1 corresponding to Tarinee Rungjirajittranon, chosen because this name is furthest up the author list.

Reply: Thank you for your suggestion. We have edited the Authors’s Contribution section according to your suggestion. (Authors’s Contribution section, line 247-251, page 10-11)

Yours faithfully,

Weerapat Owattanapanich, MD (Corresponding Author)

Division of Hematology, Department of Medicine
Faculty of Medicine, Siriraj Hospital, Mahidol University, Bangkok 10700, Thailand
Email: weerapato36733@gmail.com
Tel: +66 2 419 4448